Author's response to reviews

Title: Developing Community-based Preventive Interventions in Hong Kong: A Description of the First Phase of the Family Project

Authors:

Sunita M Stewart (sunita.stewart@utsouthwestern.edu)
Cecilia S Fabrizio (fabrizio@hku.hk)
Malia R Hirschmann (mhirsch@hku.hk)
Tai Hing Lam (hrmrlth@hku.hk)

Version: 2 Date: 24 May 2011

Author's response to reviews:

May 13, 2011

Mr. Vic Silvestre
Journal Editorial Office
BioMed Central

Subject: Changes made to article: MS: 1808541085529703
(Developing Community-based Preventive Interventions in Hong Kong: A Description of the First Phase of the Family Project)

Dear Mr. Silvestre,

Per your request, we have made the following changes to the manuscript:

• Deleted twelve blocks of quotes on pages 11 & 12
• Modified discussions of c(ii) on page 11, to replace the context of the quotes deleted

Thank you for your consideration.

Regards,
Cecilia Fabrizio